RESEARCH





Birth by caesarean section and semen quality in adulthood: a Danish population-based cohort study

Kun Huang^{1,2,3,4,5*}, Anne Gaml-Sørensen⁵, Nis Brix^{5,6}, Andreas Ernst^{5,7}, Linn Håkonsen Arendt^{5,8}, Jens Peter Ellekilde Bonde^{10,9}, Karin Sørig Hougaard^{10,11}, Gunnar Toft¹², Sandra Søgaard Tøttenborg^{10,9} and Cecilia Høst Ramlau-Hansen⁵

Abstract

Background The caesarean section (CS) rate has increased worldwide and there is an increasing public and scientific interest in the potential long-term health consequences for the offspring. CS is related to persistent aberrant microbiota colonization in the offspring, which may negatively interfere with sex hormone homeostasis and thus potentially affect the reproductive health. It remains unknown whether adult sons' semen quality is affected by CS. We hypothesize that CS is associated with lower semen guality.

Methods This study was based on the Fetal Programming of Semen Quality cohort (FEPOS, enrolled from 2017 to 2019) nested within the Danish National Birth Cohort (DNBC, enrolled from 1996 to 2002). A total of 5697 adult sons of mothers from the DNBC were invited to the FEPOS cohort, and 1044 young men participated in this study. Information on mode of delivery was extracted from the Danish Medical Birth Registry, and included vaginal delivery, elective CS before labor, emergency CS during labor and unspecified CS. The young men provided a semen sample for analysis of semen volume, sperm concentration, motility and morphology. Negative binomial regression models were applied to examine the association between CS and semen characteristics with estimation of relative differences in percentages with 95% confidence intervals (Cls).

Results Among included sons, 132 (13%) were born by CS. We found a slightly lower non-progressive sperm motility (reflecting higher progressive sperm motility) among sons born by CS compared to sons born by vaginal delivery [relative difference (95% CI): -7.5% (-14.1% to -0.4%)]. No differences were observed for other sperm characteristics. When CS was further classified into elective CS, emergency CS and unspecified CS in a sensitivity analysis, no significant differences in non-progressive motility were observed among sons born by any of the three types of CS compared to sons born vaginally.

Conclusions This large population-based cohort study found no significant evidence for an adverse effect on semen quality in adult sons born by CS.

Keywords Caesarean Section, Cohort Studies, Delivery, Obstetric, Fertility, Semen analysis

*Correspondence: Kun Huang wuweihk8028@163.com Full list of author information is available at the end of the article



© The Author(s) 2024. Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Plain language summary

Caesarean section is one of the most frequently used interventions during childbirth and global cesarean delivery rates continue to increase. The rising cesarean delivery rate has been reported to be related with series of adverse health outcomes in children, such as asthma, allergies, obesity, diabetes and even poor emotional, behavioral and educational outcomes. Still, it remains unknown whether children's reproductive health is affected by this delivery mode.

Based on data from the Fetal Programming of Semen Quality cohort (FEPOS,) nested within the Danish National Birth Cohort, we have therefore analyzed the potential effect of caesarean section on son's semen quality in 1044 young men. We found a slightly higher progressive sperm motility among sons born by caesarean section compared to sons born by vaginal delivery. No differences, however, were observed for semen volume, sperm concentration and morphology between the two delivery modes.

The FEPOS cohort is the largest population-based male offspring cohort worldwide. This is the first study aiming to examine the association between caesarean section and semen quality in adulthood. Although the findings need to be confirmed in other studies, it is reassuring that this large population-based cohort study finds no significant evidence for an adverse effect on semen quality in adult sons born by caesarean section.

Introduction

Worldwide, the caesarean section (CS) rate has almost doubled from 12% of all births in 2012 to 21% in 2015 [1] with large geographical differences. According to the World Health Organization, 6.2 million CSs are performed yearly without medical indication [2]. Understanding the long-term effects of CS on child health is important for guiding decision making for clinicians, policy makers and the parents. Potential consequences for the children's respiratory, metabolic, cardiovascular and immune function affected by CS are described [3]. However, it remains unclear whether also the reproductive health of children is affected by CS.

There is scientific and public concern about the state of male reproductive health in Western countries [4]. Poor semen quality is a major cause of infertility [5], which affects psychosocial well-being of couples worldwide [6] and has major consequences for society, due to cost of treatment, absenteeism from work and, most importantly, lower birth rates [7, 8]. Whether mode of delivery impacts semen quality remains to be investigated. It has been indicated that infants born by CS may be colonized by aberrant microbiota [9–11] that may persist into adulthood [12]. Gut microbiota dysbiosis has been suggested to negatively modulate gonadal sex hormones production and regulation and thus potentially decrease reproductive fitness [13, 14]. In particular, it has been reported that gut microbiota in the infant born via elective CS was less rich and diverse than those born by emergency CS [15, 16], as may be due to "partial" microbial exposure during emergency CS following membrane rupture [15]. To the best of our knowledge, semen quality of sons delivered by CS has not previously been studied.

In this present population-based cohort study, we aimed to examine the potential association between

birth by CS and the adult son's semen quality, under the hypothesis that CS, in general, and elective CS, in particular, is associated with lower semen quality.

Materials and methods Participants

This study is based on the Fetal Programming of Semen Quality (FEPOS) cohort [17], nested within the Danish National Birth Cohort (DNBC) [18]. In the DNBC, from 1996 to 2002, pregnant women were enrolled and interviewed by computer-assisted telephone interviews conducted around 16 and 30 weeks' gestation, and at 6 months after childbirth.

The participants in the FEPOS cohort were recruited from March 2017 to December 2019. Recruitment to FEPOS is described in detail elsewhere [17]. In short, sons were eligible for invitation if their mothers responded to the first two telephone interviews during pregnancy in the DNBC and provided a gestational blood sample for the biobank within the DNBC. Invitation to the FEPOS cohort was further restricted to the following criteria: Son of at least 18 years and 9 months of age, living in close proximity to the FEPOS clinics in Copenhagen (the capital and largest city in Denmark) or Aarhus (the second largest city in Denmark). Sons were encouraged to decline participation if they had undergone sterilization, cancer treatment, orchidectomy, or had one or no testicles in the scrotum.

During the study period, 5697 of the 21,623 eligible young men were randomly invited to participate in FEPOS. Of these, 1174 (21%) responded to an online questionnaire on health behaviors, whereafter 1058 delivered a semen sample (participation rate 19%). After excluding sons with no semen sample and/or missing data on delivery mode, 1044 mother-son pairs (18%) were included in our main analysis (Fig. 1).

Delivery mode

Information on delivery mode was extracted from the Danish Medical Birth Registry (DMBR) [19, 20] based on ICD-10 diagnostic codes and procedure codes according to the Nordic Medico-Statistical Committee classification of surgical procedures. Delivery mode included vaginal delivery (ICD-10: DO800, DO801, DO808, DO809, DO810, DO813, DO814, DO815, DO840, DO841), elective CS before labor (ICD-10: DO820 and operation codes KMCA10B or KMCA11), emergency CS during labor (ICD-10: DO821, DO842 and operation codes KMCA10D, KMCA10E) and unspecified CS (ICD-10: DO828, DO829 and operation codes KMCA10A). The latter three categories of CS (elective CS, emergency CS and unspecified CS) were merged into an overall CS exposure category in the main analysis due to low numbers in some of the separate categories.

Semen quality characteristics

Participants provided a semen sample for analysis either at the clinic or at home. If collected at home, participants were recommended to keep the sample warm during transportation to the clinic and deliver the sample within one hour of ejaculation. Recommended abstinence time was 2–3 days.

Collection and analysis of semen followed recommendations by the WHO 2010 [21]. Analysis was conducted by two trained biomedical laboratory technicians and started within a maximum of two hours from ejaculation in 1029 out of 1044 semen samples (99%). In total, 786 (75%) semen samples were analyzed within one hour. Semen volume was measured by weighing of the sample in a pre-weighed container assuming 1 g = 1 ml. After liquefaction at 37 °C, sperm concentration was determined on two aliquots of semen using an Improved Neubauer Hemocytometer. Total sperm count was calculated by multiplication of semen volume and concentration. Sperm motility was determined by assessing progressive, non-progressive and immotile spermatozoa for at least 200 spermatozoa within each of two fresh drops of semen. The outcome used in the study was progressive



Fig. 1 Flow chart of participants, the FEPOS Cohort, Denmark, 1998–2019

motility, which, to ensure optimal model fit, was assessed as non-progressive motility modeled as non-progressive+immotile spermatozoa in percentage. Morphology was analyzed at the Reproductive Medicine Centre, Skåne University Hospital, in Malmö, Sweden. The percentage of normal spermatozoa was assessed in approximately 200 spermatozoa per slide.

External quality control with the European Society of Human Reproduction and Embryology (ESHRE) External Quality Assessment scheme (Centre for Andrology, Karolinska University Hospital, Stockholm, Sweden) indicated no distinct differences between the FEPOS laboratory technologists and the expert reference examiners [17].

Covariates

Potential confounders were identified a priori using existing literature and Directed Acyclic Graphs (DAGs) [22] (Additional file 1: Appendix S1). Maternal age at delivery was retrieved from DMBR. Information on maternal smoking in the first trimester, pre-pregnancy body mass index (BMI) and highest socioeconomic status of the parents was provided by the mothers in the first DNBC interview. Socioeconomic status was defined according to the International Standard Class of Occupation and Education codes (ISCO-88 and ISCED). Categorization of the covariates was presented in Table 1.

Information on precision variables was recorded at the clinical visit, including place of semen sample collection (at home or in the clinic), abstinence time (in days), spillage of semen sample (no; yes), and interval from ejaculation to analysis (in minutes).

Information on pregnancy complications, labor induction, gestational age at birth (in weeks) and birth weight (in grams) were used in sensitivity analyses. Pregnancy complications were self-reported in the women's interview around 16 and 30 weeks of gestation and 6 months after childbirth and included diabetes mellitus (type 1, type 2 or gestational diabetes), hypertension and preeclampsia. Women, who reported one or several of these conditions, were defined as having pregnancy complications (none; one or more). Information on gestational age at birth, birth weight, and labor induction (pharmaceutical or non-pharmaceutical based on ICD diagnose codes DO837 and DO847 and procedure codes KMAC00, KMAC96A, BKHD20, BKHD20A and BKHD 21) were obtained from DMBR.

Statistical analysis

Data management and statistical analyses were conducted with STATA MP, version 15.1 (StataCorp, College Station, TX). According to the requirements of regulations [GDPR, Regulation (EU), 2016/679 of 25 May 2018)], a calculated percentile must be based on at least five observations. Thus, pseudo percentile 5, 50 (median), and 95 for the semen characteristics were calculated by delivery mode using STATA's -sumat- commands.

As the distributions of the semen characteristics were over-dispersed, we applied negative binomial regression models to examine the association between CS and semen characteristics using STATA's -nbreg- package. We estimated crude and adjusted relative differences in percentages with 95% confidence intervals (CIs) for each semen characteristics by comparing sons born by CS to those born by vaginal delivery.

Azoospermic men were excluded from analyses of motility and morphology (n=17), and participants reporting spillage of the semen sample were excluded from analyses of semen volume and total sperm count (n=180). All models were adjusted for abovementioned selected confounders, place of semen sample collection and abstinence time. The analysis of sperm concentration and morphology was further adjusted for spillage, and the analysis of motility was further adjusted for interval from ejaculation as sperm motility decreases with time [23].

We applied selection weights to all analyses to account for non-participation [24]. These were estimated as the inverse probability of participation derived from a multivariable logistic regression with participation status (yes; no) as the dependent variable and the primary exposure variable (delivery mode) and the potential confounding factors, in addition to region (Aarhus; Copenhagen), parental time to pregnancy (TTP) including use of medically assisted reproduction (MAR), included as explanatory variables. TTP and MAR were considered indicators of parental fertility, and these data were obtained in the first interview in the DNBC.

Assumptions were checked for each model. We compared the observed distribution of each semen characteristic against the model-based distributions from the fitted model using QQ-plots. Furthermore, the standardized deviance residuals were plotted against the model-based predictions. The model check was compatible with the assumptions (data not shown).

We performed three sensitivity analyses to assess confounding by indication. First, we further adjusted for birth weight z-scores [25] as being small or large for gestational age might be an important indicator for CS, and birth weight for gestational age has been reported to be associated with semen quality [26, 27]. Birth weight z-score was not included in the main analyses due to the risk of collider-stratification bias [28, 29]. Second, we restricted our analyses to women without induction of labor and pregnancy complications. Labor induction has been suggested by some researchers Table 1 Baseline characteristics by delivery mode, N = 1044, the FEPOS cohort, Denmark, 1998–2019

Background characteristics	Delivery mode	Missing	
	Vaginal delivery (n=912)	CS (n = 132)	n (%)
Maternal characteristics			
Maternal age at delivery, years, mean \pm SD	30.9 ± 0.1	31.4±0.3	0 (0.0)
Highest socioeconomic status of parents, n (%)			0 (0.0)
High-grade professional	307 (33.7)	50 (37.9)	
Low-grade professional	307 (33.7)	40 (30.3)	
Skilled worker	171 (18.8)	25 (18.9)	
Unskilled worker	84 (9.2)	11 (8.3)	
Student, economically inactive and un-definable	43 (4.7)	6 (4.6)	
Pre-pregnancy BMI, kg/m ² , n (%)			25 (2.4)
<18.5	55 (6.2)	6 (4.7)	
18.5–24.9	663 (74.4)	85 (66.4)	
25.0–29.9	136 (15.3)	27 (21.1)	
≥30.0	37 (4.2)	10 (7.8)	
Smoking in the 1st trimester, n (%)			<15 (<1.4) ^b
Nonsmoker	702 (77.7)	101 (78.9)	
1–10 daily cigarettes	171 (18.9)	<25 (<18.9) ^b	
> 10 daily cigarettes	31 (3.4)	< 5 (< 3.8) ^b	
Pregnancy complications, n (%) ^a			0 (0.0)
None	860 (94.3)	120 (90.9)	
One or more	52 (5.7)	12 (9.1)	
Son's characteristics			
Gestational week at delivery, mean \pm SD	39.8±1.5	39.3±2.1	0 (0.0)
Birth weight, kg, mean \pm SD	3.7±0.5	3.6±0.7	12 (1.1)
Semen-related characteristics			
Place of semen sample delivery, n (%)			6 (0.6)
At home	125 (13.7)	13 (9.9)	
In the clinic	781 (85.6)	119 (90.2)	
Abstinence time, days, mean \pm SD	2.3 ± 0.0	2.4±0.1	<5 (<0.5) ^b
Interval from ejaculation to analysis, minutes, mean \pm SD	49.9±0.7	50.2 ± 1.6	8 (0.8)
Spillage of semen sample, n (%)			5 (0.5)
No	747 (82.3)	112 (85.5)	
Yes	161 (17.7)	19 (14.5)	

SD standard deviation

Body mass index (BMI) = weight(kg)/height(m)²

^a Pregnancy complications defined as women reporting one or more of the following conditions: diabetes mellitus (type 1, type 2 or gestational diabetes), hypertension and preeclampsia

^b Due to data regulations that numbers less than five are not allowed to report, the numbers in the table have been changed or combined to cover up the numbers less than five

as an independent risk factor for CS [30, 31], and the indications for induction may be the same as for CS. In addition, severe pregnancy complications might be a potential medical indicator for CS. Third, CS was further classified into elective CS, emergency CS and unspecified CS, with vaginally-born sons as reference, to examine whether the type of CS was important for semen quality.

Results

Among the 1044 sons included, 132 (13%) were born by CS. Mothers giving birth by CS were on average older, had higher pre-pregnancy BMI and experienced more pregnancy complications than mothers giving birth vaginally (Table 1).

Overall, sons born by CS had similar median values for the assessed semen characteristics compared to sons born vaginally (Table 2).

In the main analysis, we found a slightly lower percentage of non-progressive sperm motility (reflecting higher percentage of progressive sperm motility) for sons born by CS than for sons born vaginally [relative difference (95% CI): -7.5% (-14.1% to -0.4%)]. No differences were observed for other sperm characteristics (Table 3).

In the sensitivity analyses, with further adjustment for birth weight z-scores (model 1), or restriction to women without labor induction and pregnancy complications (model 2), the results did not change essentially. In model 3, when CS was further classified into elective CS, emergency CS and unspecified CS, no significant differences in non-progressive motility or any of the other semen characteristics were observed among sons born by any

Discussion

To the best of our knowledge, this is the first study to examine the association between CS and semen quality in sons at adulthood. Overall, there was no significant evidence for an adverse effect of CS on semen quality.

There is no clear biological explanation for the slightly higher percentage of progressively motile spermatozoa among the young adults born by CS compared to sons born vaginally. Nevertheless, the effect size was small, and no associations were observed between CS and the other semen characteristics. Further, we did not observe associations when taking the different types of CS into account in sensitivity analysis. Therefore, the current result is likely a chance finding.

 Table 2
 Semen characteristics ^a by delivery mode, the FEPOS cohort, Denmark, 1998–2019

Semen parameters	Number of observations	Vaginal delivery			Cesarea	Cesarean section		
		P ₅	Median	P ₉₅	P ₅	Median	P ₉₅	
Semen volume, mL ^b	858	1.0	2.7	5.4	1.0	2.9	5.9	
Total sperm count, million ^b	858	7.6	101.2	413.6	8.8	105.7	398.1	
Sperm concentration, million/mL	1042	2.8	37.9	136.5	1.5	41.3	149.3	
Progressive sperm motility, % ^c	1025	29.2	63.0	83.2	35.4	65.5	82.8	
Morphologically normal sperm, % ^c	1020	0.5	6.0	15.0	0.1	6.0	14.5	

^a Pseudo percentiles were presented and were calculated based on the average of five values

^b 180 samples were excluded due to spillage

^c Progressive motility and morphology were not available for 17 samples due to azoospermia

Table 3	Relative differences in	n percentage in semen d	characteristics by delive	ry mode, the FEPOS cohort	, Denmark, 1998–2019
				· ·	

Semen parameters	Number of observations ^a	Relative differences in percentage ^b		
		Unadjusted model	Adjusted model	
		%	% (95% Cl)	
Semen volume ^f	818	5.6 (- 6.8 to 18.5)	5.3 (– 5.3 to 17.1) ^c	
Total sperm count ^f	818	0.9 (- 17.4 to 22.3)	- 0.8 (- 14.9 to 15.8) ^c	
Sperm concentration	988	5.4 (- 11.7 to 24.2)	4.3 (- 11.2 to 22.5) ^d	
Non-progressive motility ^{gh}	965	- 6.0 (- 13.8 to 1.3)	- 7.5 (- 14.1 to - 0.4) ^e	
Morphologically normal sperm ^g	966	2.0 (- 11.5 to 17.4)	– 2.0 (– 14.2 to 11.8) ^d	

^a Number of observations in adjusted model for each parameters

^b Relative percentage differences were obtained using sons born of vaginal delivery as the reference

^c Adjusted for maternal age at delivery, highest socioeconomic status of parents, maternal pre-pregnancy BMI, maternal smoking in the 1st trimester, place of semen sample collection, abstinence time

^d Further adjusted for spillage

^e Further adjusted for spillage and interval from ejaculation to analysis

^f 180 samples were excluded for the analyses on semen volume and total sperm count due to spillage

⁹ For non-progressive motility and morphology, 17 samples were not available for analyses due to azoospermia

^h Non-progressive motility was modeled as (non-progressive + immotile spermatozoa) %, and a negative estimate indicated a relatively lower non-progressive motility (ie. higher progressive motility)

Semen parameters	Analysis strategies	Number of observations	Delivery mode	Relative differences in percentage		
				Unadjusted model	Adjusted model ^a	
				%	% (95% CI)	
Semen volume ^b	Model 1 ^f	806	Overall CS	5.6 (- 6.8 to 18.5)	5.4 (- 5.4 to 17.5)	
	Model 2 g	691	Overall CS	5.5 (- 7.0 to 19.7)	5.2 (- 6.4 to 18.3)	
	Model 3 h	818	Elective CS (n = 35)	– 4.7 (– 21.8 to 16.0)	– 2.4 (– 16.5 to 14.1)	
			Emergency CS (n = 52)	10.9 (- 4.8 to 29.2)	10.1 (- 6.7 to 30.0)	
			Unspecified CS ($n = 17$)	11.1 (– 14.5 to 44.5)	7.8 (– 9.9 to 29.0)	
Total sperm count ^b	Model 1 ^f	806	Overall CS	0.9 (- 17.4 to 22.3)	– 0.5 (– 14.8 to 16.1)	
	Model 2 g	691	Overall CS	– 0.4 (– 19.7 to 23.5)	0.0 (- 15.9 to 19.0)	
	Model 3 h	818	Elective CS (n = 35)	- 20.0 (- 41.7 to 9.7)	- 16.4 (- 36.7 to 10.5)	
			Emergency CS (n = 52)	6.0 (– 18.5 to 37.8)	7.2 (- 12.0 to 31.0)	
			Unspecified CS ($n = 17$)	29.4 (– 17.7 to 103.3)	9.5 (- 22.2 to 54.0)	
Sperm concentration ^c	Model 1 ^f	975	Overall CS	5.4 (- 11.7 to 24.2)	4.6 (- 11.2 to 23.3)	
	Model 2 g	828	Overall CS	3.9 (- 13.4 to 24.4)	2.5 (- 14.0 to 22.3)	
	Model 3 h	988	Elective CS (n=41)	– 7.9 (– 29.1 to 21.0)	- 4.5 (- 27.6 to 26.0)	
			Emergency CS (n = 60)	13.1 (- 9.6 to 41.3)	10.8 (- 11.3 to 38.5)	
			Unspecified CS ($n = 20$)	8.7 (- 26.4 to 60.4)	4.1 (- 23.1 to 41.0)	
Non– progressive motility ^{de}	Model 1 ^f	952	Overall CS	– 6.0 (– 13.8 to 1.3)	- 7.3 (- 13.0 to 0.0)	
	Model 2 g	812	Overall CS	– 5.4 (– 12.1 to 2.6)	– 6.8 (– 14.3 to 1.2)	
	Model 3 h	965	Elective CS (n = 38)	– 9.0 (– 19.8 to 3.2)	– 10.8 (– 21.3 to 1.1)	
			Emergency CS (n = 58)	– 6.7 (– 15.8 to 3.3)	- 7.4 (- 16.0 to 1.2)	
			Unspecified CS (n = 20)	1.9 (- 14.6 to 21.3)	– 1.1 (– 16.8 to 17.9)	
Morphologically normal sperm ^{ce}	Model 1 ^f	953	Overall CS	2.0 (- 11.5 to 17.4)	- 2.1 (- 14.4 to 12.0)	
	Model 2 g	812	Overall CS	1.6 (- 12.8 to 18.3)	– 2.0 (– 15.3 to 13.4)	
	Model 3 h	966	Elective CS (n = 39)	– 19.4 (– 36.7 to 2.7)	- 20.4 (- 39.8 to 5.3)	
			Emergency CS (n = 58)	11.4 (- 7.8 to 34.6)	5.6 (– 9.1 to 22.7)	
			Unspecified CS ($n = 20$)	15.4 (- 16.5 to 59.4)	14.3 (- 11.9 to 48.4)	

Table 4 Sensitive analyses for relative differences in percentage in semen characteristics by delivery mode, the FEPOS cohort, Denmark, 1998–2019

CS caesarean section

^a Basic set of variables that were adjusted for all semen parameters included maternal age at delivery, highest socioeconomic status of parents, maternal pre-

pregnancy BMI, maternal smoking in the 1st trimester, place of semen sample collection and abstinence time

^b For the analyses of semen volume and total sperm count, 180 samples were excluded due to spillage

^c For the analysis of sperm concentration and morphologically normal sperm, spillage was further adjusted for

^d Non-progressive motility was modeled as (non-progressive + immotile spermatozoa) %, and a negative estimate indicated a relatively lower non-progressive motility (ie. higher progressive motility). In the analysis, spillage and internal from ejaculation to analysis were further adjusted for

^e For non-progressive motility and morphology, 17 samples were not available for analyses due to azoospermia

^f Model 1: Birth weight z-scores were further adjusted for

^g Model 2: Restricted to women without labor induction and pregnancy complications

^h Model 3: CS was further classified into elective CS, emergency CS and unspecified CS, and percentage differences were obtained by comparing with the sons born of vaginal delivery

This study has several strengths. The FEPOS cohort is the largest population-based male offspring cohort worldwide with detailed and prospectively collected data on maternal health in pregnancy, including information on birth outcomes, which allowed for comprehensive confounding control and reduced the risk of recall bias. Besides the potential confounders, precision variables were adjusted for in the regression models to increase the statistical precision. As the prevalence of urogenital diseases was low in this population and are not suspected to affect the results, these diseases were not adjusted for in the analyses. Information on CS was prospectively registered, and data from the DMBR are considered valid [21] with high predictive values, although we cannot rule out some misclassification regarding the type of CS. Neither participants, nor the biomedical laboratory scientists conducting the semen quality analyses, were aware of the objective of the current study. Therefore, the risk of differential misclassification is limited.

Some limitations must be considered. The participation rate in FEPOS was 19%, which increases the risk of selection bias. However, we consider participation not to be related to CS (CS rate was 13% and 14% in participants and non-participants, respectively), and that self-selection into the cohort did not depend on semen quality. The sons were young and most would not yet be aware of their fertility status, which minimizes the risk of self-selection. However, sons from families with fertility problems could be more likely to participate. Therefore, we estimated selection weights, including parental time to pregnancy and MAR use, to take the potential selective participation into account. Further, in this study, we could not differentiate between indicators for elective CS. Medical indication for elective CS (e.g. preeclampsia) could interfere with the development of the male reproductive organs [32]. The DMBR have only registrations of elective CS on maternal request from late 2001 and onwards. We could therefore not differentiate between reasons for elective CS to examine if potential effects associated with CS could be attributed to the underlying medical conditions or the surgical procedure. We performed sensitivity analysis to assess confounding by indication by adjusting for birth weight for gestational age, excluding women with induction of labor and pregnancy complications and categorizing CS into elective, emergency and unspecified CS. Still, other sources of unmeasured confounding, in particular confounding by indication, cannot not be ruled out.

Conclusions

In summary, this large population-based cohort study found no significant evidence for an adverse effect on semen quality in adult sons born by CS.

Abbreviations

CS	Caesarean section
FEPOS	: Fetal Programming of Semen Quality
DNBC	: Danish National Birth Cohort
CI	: Confidence interval
OR	: Odds ratio

Supplementary Information

The online version contains supplementary material available at https://doi. org/10.1186/s12978-024-01761-w.

Additional file 1: Appendix S1. DAG for CS and semen quality.

Acknowledgements

The authors are grateful to all the participants and to biomedical laboratory technicians Marianne Lipka Flensborg and Joan Dideriksen for running the clinics and collecting data. We would also like to thank Josefine Rahbæk

Larsen for assisting with recruitment and data entry, Lone Fredslund and Inge Eisensee for data management, and Cecilia Tingsmark for doing the morphology analysis. The Danish National Birth Cohort was established with a significant grant from the Danish National Research Foundation. Additional support was obtained from the Danish Regional Committees, the Pharmacy Foundation, the Egmont Foundation, the March of Dimes Birth Defects Foundation, the Health Foundation and other minor grants. The DNBC Biobank has been supported by the Novo Nordisk Foundation and the Lundbeck Foundation. Support for follow-up of mothers and children has been given by the Danish Medical Research Council (SSVF 0646, 271-08-0839/06-066023, O602-010428, 0602-02738B), the Lundbeck Foundation (195/04, R100-A9193), The Innovation Fund Denmark 0603-00294B (09-067124), the Nordea Foundation (02-2013-2014), Aarhus Ideas (AU R9-A959-13-S804), University of Copenhagen Strategic Grant (IFSV 2012), and the Danish Council for Independent Research (DFF – 4183-00594 and DFF – 4183-00152).

Author contributions

KH and CHRH conceptualized and designed the study. KH and AG-S performed the statistical analyses and wrote the first draft in close collaboration with NB, AE and LHA. SST and JPEB acquired the funding, and GT, SST, KSH, JPEB and CHR-H planned and headed the data collection for FEPOS. All authors reviewed the article critically and approved the final version to be published.

Funding

This publication is part of the ReproUnion collaborative study, co-financed by the European Union, Intereg V ÖKS (20200407). The FEPOS project was further funded by the Lundbeck Foundation (R170-2014-855), the Capital Region of Denmark, Medical doctor Sofus Carl Emil Friis and spouse Olga Doris Friis's Grant, Axel Muusfeldt's Foundation (2016–491), A.P. Møller Foundation (16–37), the Health Foundation and Dagmar Marshall's Fond. This work has also been funded by the National Natural Science Foundation of China (82273639) and National Key R&D Program of China (2022YFC2702901).

Availability of data and materials

The dataset analyzed in the study is not publicly available due to national data security legislation on sensitive personal data. Access to data can be applied for at dnbc-research@ssi.dk.

Declarations

Ethics approval and consent to participate

The data collection within the DNBC was approved by The Committee for Biomedical Research Ethics in Denmark [(KF)01-471/94]. The establishment of the FEPOS cohort was approved by the Scientific Research Ethics Committee for Copenhagen and Frederiksberg (No. H-16015857). This study was approved by the steering committee of the DNBC (Ref. no. 2019–21) and registered at the Danish Data Protection Agency 2015-57-0002 (1150). Each participant provided written informed consent. We maintained participants' privacy and confidentiality throughout the participant's enrollment into the study and data collection and analysis.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Department of Maternal, Child and Adolescent Health, School of Public Health, Anhui Medical University, No 81 Meishan Road, Hefei 230032, China. ²MOE Key Laboratory of Population Health Across Life Cycle, Hefei 230032, China. ³NHC Key Laboratory of Study On Abnormal Gametes and Reproductive Tract, Hefei 230032, China. ⁴Anhui Provincial Key Laboratory of Environment and Population Health Across the Life Course, Hefei 230032, China. ⁵Department of Public Health, Research Unit for Epidemiology, Aarhus University, 8000 Aarhus C, Denmark. ⁶Department of Clinical Genetics, Aarhus University Hospital, 8200 Aarhus N, Denmark. ⁸Department of Obstetrics and Gynecology, Aarhus University Hospital, 8200 Aarhus N, Denmark. ⁹Department of Occupational and Environmental Medicine, Bispebjerg and Frederiksberg Hospital, University of Copenhagen, 2400 Copenhagen, NV, Denmark. ¹⁰Department of Public Health, University of Copenhagen, 1353 Copenhagen K, Denmark. ¹¹National Research Centre for the Working Environment, 2100 Copenhagen, OE, Denmark. ¹²Steno Diabetes Center Aarhus, Aarhus University Hospital, 8200 Arhus N, Denmark.

Received: 22 February 2022 Accepted: 23 February 2024 Published online: 08 March 2024

References

- Boerma T, Ronsmans C, Melesse DY, Barros AJD, Barros FC, Juan L, et al. Global epidemiology of use of and disparities in caesarean sections. Lancet. 2018;392:1341–8.
- Gibbons L, Belizán JM, Lauer JA, Betrán AP, Merialdi M, Althabe F, for WHO. The global numbers and costs of additionally needed and unnecessary caesarean sections performed per year: overuse as a barrier to universal coverage. 2010.
- Sandall J, Tribe RM, Avery L, Mola G, Visser GH, Homer CS, et al. Short-term and long-term effects of caesarean section on the health of women and children. Lancet. 2018;392:1349–57.
- Skakkebaek NE, Rajpert-De Meyts E, Buck Louis GM, Toppari J, Andersson AM, Eisenberg ML, et al. Male Reproductive Disorders and Fertility Trends: Influences of Environment and Genetic Susceptibility. Physiol Rev. 2016;96(1):55–97.
- Levine H, Jørgensen N, Martino-Andrade A, Mendiola J, Weksler-Derri D, Mindlis I, et al. Temporal trends in sperm count: a systematic review and meta-regression analysis. Hum Reprod Update. 2017;23:646–59.
- Vander Borght M, Wyns C. Fertility and infertility: definition and epidemiology. Clin Biochem. 2018;62:2–10.
- Baird DT, Collins J, Egozcue J, Evers LH, Gianaroli L, Leridon H, et al. Fertility and ageing. Hum Reprod Update. 2005;11:261–76.
- Christiansen T, Erb K, Rizvanovic A, Ziebe S, Mikkelsen Englund AL, Hald F, et al. Costs of medically assisted reproduction treatment at specialized fertility clinics in the Danish public health care system: results from a 5-year follow-up cohort study. Acta Obstet Gynecol Scand. 2014;93:64–72.
- Jakobsson HE, Abrahamsson TR, Jenmalm MC, Harris K, Quince C, Jernberg C, et al. Decreased gut microbiota diversity, delayed Bacteroidetes colonisation and reduced Th1 responses in infants delivered by caesarean section. Gut. 2014;63:559–66.
- Shao Y, Forster SC, Tsaliki E, Vervier K, Strang A, Simpson N, et al. Stunted microbiota and opportunistic pathogen colonization in caesareansection birth. Nature. 2019;574(7776):117–21.
- Giannattasio A, Guarino A. Caesarean-section and neonatal gut microbiome: short and long term effects and new targets for early prevention. Ann Nutr Metab. 2018;73:1–3.
- 12. Goedert JJ. Intestinal microbiota and health of adults who were born by cesarean delivery. JAMA Pediatr. 2016;170:1027.
- Markle JG, Frank DN, Mortin-Toth S, Robertson CE, Feazel LM, Rolle-Kampczyk U, et al. Sex differences in the gut microbiome drive hormonedependent regulation of autoimmunity. Science. 2013;339:1084–8.
- 14. Silva MSB, Giacobini P. Don't trust your gut: when gut microbiota disrupt fertility. Cell Metab. 2019;30:616–8.
- Azad MB, Konya T, Maughan H, Guttman DS, Field CJ, Chari RS, et al. Gut microbiota of healthy Canadian infants: profiles by mode of delivery and infant diet at 4 months. CMAJ. 2013;185:385–94.
- Chu DM, Ma J, Prince AL, Antony KM, Seferovic MD, Aagaard KM. Maturation of the infant microbiome community structure and function across multiple body sites and in relation to mode of delivery. Nat Med. 2017;23:314–26.
- Keglberg Hærvig K, Bonde JP, Ramlau-Hansen CH, Toft G, Hougaard KS, Specht IO, et al. Fetal programming of semen quality (FEPOS) cohort—a DNBC male-offspring cohort. Clin Epidemiol. 2020;12:757–70.
- Olsen J, Melbye M, Olsen SF, Sørensen TI, Aaby P, Andersen AM, et al. The Danish National Birth Cohort–its background, structure and aim. Scand J Public Health. 2001;29:300–7.
- Knudsen LB, Olsen J. The Danish Medical Birth registry. Dan Med Bull. 1998;45(3):320–3.

- Bliddal M, Broe A, Pottegård A, Olsen J, Langhoff-Roos J. The Danish Medical Birth register. Eur J Epidemiol. 2018;33:27–36.
- 21. WHO. Examination and processing of human semen. Vol. Edition, V, World Health. 2010.
- 22. Greenland S, Pearl J, Robins JM. Causal diagrams for epidemiologic research. Epidemiology. 1999;10:37–48.
- Makler A, Zaidise I, Paldi E, Brandes JM. Factors affecting sperm motility. I. In vitro change in motility with time after ejaculation. Fertil Steril. 1979;31:147–54.
- 24. Hernán MA, Hernández-Díaz S, Robins JM. A structural approach to selection bias. Epidemiology. 2004;15:615–25.
- Data Analysis. Probability theory. In: Kreyszig E, eds. Advanced engineering mathematics. 10th ed. Wiley; 2011: 1014.
- Kahn LG, Widen EM, Janevic T, Straka N, Liu X, Cirillo PM, et al. The relation of birth weight and adiposity across the life course to semen quality in middle age. Epidemiology. 2019;30(Suppl 2):S17–27.
- 27. Thorsted A, Lauridsen J, Høyer B, Arendt LH, Bech B, Toft G, et al. Birth weight for gestational age and the risk of infertility: a Danish cohort study. Hum Reprod. 2020;35:195–202.
- Hernández-Díaz S, Schisterman EF, Hernán MA. The birth weight "paradox" uncovered? Am J Epidemiol. 2006;164:1115–20.
- Wilcox AJ, Weinberg CR, Basso O. On the pitfalls of adjusting for gestational age at birth. Am J Epidemiol. 2011;174:1062–8.
- Thaens A, Bonnaerens A, Martens G, Mesens T, Van Holsbeke C, De Jonge E, et al. Understanding rising caesarean section trends: relevance of inductions and prelabour obstetric interventions at term. Facts Views Vis Obgyn. 2011;3:286–91.
- Danilack VA, Dore DD, Triche EW, Muri JH, Phipps MG, Savitz DA. The effect of labour induction on the risk of caesarean delivery: using propensity scores to control confounding by indication. BJOG. 2016;123:1521–9.
- Szczuko M, Kikut J, Komorniak N, Bilicki J, Celewicz Z, Ziętek M. The role of arachidonic and linoleic acid derivatives in pathological pregnancies and the human reproduction process. Int J Mol Sci. 2020;21:9628.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.